Author’s response to reviews

Title: Severe Hypoglycemia Reduces the Shivering Threshold in Rabbits

Authors:

Keiichi Wada (kwada07@gmail.com;kwada@yamanashi.ac.jp)
Taishi Masamune (ezn00202@nifty.ne.jp)
Hirofumi Ino (inohirofumi19770626@yahoo.co.jp)
Kenta Ueda (uedak@yamanashi.ac.jp)
Tadahiko Ishiyama (ishiyama@yamanashi.ac.jp)
Daniel Sessler (DS@OR.org)
Takashi Matsukawa (tamashim@yamanashi.ac.jp)

Version: 3 Date: 25 Jun 2019

Author’s response to reviews:

Editor

We appreciate the technical comments. The followings are our point-by-point responses:

1. In Funding, please state whether or not the funding body played any roles in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: We added the sentence (line 25-26, page 1).

2. The Authors Contributions section should specify the individual contributions of each author. Each author should also be specifically mentioned by their unique initials.

Response: We revised the sentence (line 30-32, page 1).
3. Please ensure that your work follows the ARRIVE guidelines (https://www.nc3rs.org.uk/arrive-guidelines).

Response: We checked the manuscript according to the guidelines.

4. We note that the current submission contains some textual overlap with other previously published works, in particular:


This overlap mainly exists in the Methods section. While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please re-phrase this section to minimise overlap.

Response: We rewrote some sentences in Methods section.

5. Upon resubmission, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please ensure the document is in the final form for publication; please upload only files that are to be published and remove any files that should not form part of the final manuscript (e.g. covering letters, responses to reviewers, etc). If you wish to respond to these revision requests, please insert the information in to the designated input box only.

Response: We cleaned the manuscript and removed unnecessary files.